Author's response to reviews

Title: Trans-visceral migration of retained surgical gauze as a cause of intestinal obstruction: a case report

Authors:

Nello Grassi (NG40147@policlinico.pa.it)
Calogero Cipolla (calogerocipolla@virgilio.it)
Adriana Torcivia (adrianatorcivia@libero.it)
Alessandro Bottino (AB40613@policlinico.pa.it)
Eugenio Fiorentino (efiorentino@unipa.it)
Leonardo Ficano (leoficano@tin.it)
Gianni Pantuso (gianpant@katamail.com)

Version: 7 Date: 16 January 2008

Author's response to reviews: see over
To the Editorial Staff of the
Journal of Medical Case Reports

January 16, 2008

I would like to submit the revised version of the article “Trans-visceral migration of retained surgical gauze as a cause of intestinal obstruction: a case report” again, after making the requested changes.

Kindest regards,

Calogero Cipolla MD